Author's response to reviews

Title: Ethnicity and gender related differences in extended intraesophageal pH monitoring parameters in infants: a retrospective study

Authors:

Dena Nazer (nazer@wayne.edu)
Ronald Thomas (rthomas@crcm.med.wayne.edu)
Vasundhara Tolia (vtolia@med.wayne.edu)

Version: 5 Date: 24 June 2005

Author’s response to reviews: see over
Dear Dr Moylan,
Thanks again for considering our manuscript for publication and I really appreciated all your comments. Below I have a point by point confirmation of addressing the changes that you requested. Please let me know of any other changes you may recommend. Again I appreciated all your comments and how much our manuscript has improved after addressing them.

Formatting changes requested
----------------------------------------
1. **Title and section headings** - Please remove all underlining: **Done**

2. **Author list** - Please can you write the list of authors in the manuscript exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system, and then put the full affiliations of each beneath: **Done**

**Qualifications** - Please remove the authors' qualifications (e.g. PhD) from the title page: **Done**

**Abstract** - Please include the abstract in the manuscript file - this should be the same as that on the submission system: **Done**

**Header** - Please remove the text from the top of every page: **Done**

**Manuscript sections** - Manuscript sections should include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any).

Therefore please place the figure legend in the correct position after the references but before the tables: **Done**

**Author contributions** Should be Authors’ contributions: **Done**

**Figures** - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: **Done** [http://www.biomedcentral.com/info/ifora/cropping Figure 1](http://www.biomedcentral.com/info/ifora/cropping Figure 1).

**References** - The reference list should contain all authors’ names, regardless of how many there are. The term “et al” should not be used. Please correct. **Done**

Ref 29 is also incorrectly cited - the journal is not a BMC-series journal. **Done**